Author’s response to reviews

Title: Assessment of two complementary influenza surveillance systems: Sentinel primary care Influenza-like Illness versus severe hospitalized laboratory-confirmed influenza using the moving epidemic method

Authors:

Nuria Torner (nuria.torner@gencat.cat)
Luca Basile (chags@gencat.cat)
Ana Martinez (a.martinez@gencat.cat)
Cristina Rius (crius@aspb.cat)
Pere Godoy (pere.godoy@gencat.cat)
Mireia Jane (mireia.jane@gencat.cat)
Angela Dominguez (angela.dominguez@ub.edu)

Version: 2 Date: 28 Jan 2019

Author’s response to reviews:

Dear Editor:

Thank you for the opportunity to improve this manuscript by reducing textual overlap. The entire content has been rewritten however you may still find some overlap in the Methods section because there are several articles based on data from this surveillance system.

The novelty of our study has been included for clarification in the cover letter.